Author's response to reviews

Title: Estrogen receptor alpha (ERα) mRNA copy numbers in immunohistochemically positive-, and negative breast cancer tissues

Authors:

Indira - Poola - (ipoola@howard.edu)
Qingqi - Yue - (yueqingqi@yahoo.com)

Version: 5 Date: 9 March 2007

Author's response to reviews:

March 9, 2007
To
The Editor
BMC Cancer

REF:
Manuscript # 1961808621131316

Title: Estrogen receptor alpha (ERα) mRNA copy numbers in immunohistochemically positive-, and negative breast cancer tissues

Authors: I. Poola and Q. Yue

Dear Editor,
This is with reference to email dated 8th of March asking to include limitations of the approach and a cautionary note in the discussion. In response to that we have included limitations and a cautionary note. Please see the last paragraph of the discussion section.

In addition, the stated editorial changes are also made. Specifically, we added the name of the IRB in methods, changed the verbiage of the competing interests and included a separate section on author contributions between acknowledgements and competing interests. We have also cropped the figures as far as we could. Unfortunately, further cropping is not feasible due to limitations of the program used for generating the figures.

Finally, you have our approval to make any minor changes if required.

We hope this manuscript will proceed for publication at this stage.

Sincerely,

Indira Poola, Ph.D
Professor of Biochemistry and Molecular Biology